Author's response to reviews

Title: Development and tracking of central patterns of subcutaneous fat of rural South African children aged 5 to 15 years: Ellisras Longitudinal Study

Authors:

Kotsedi D Monyeki (dmonyeki@yahoo.com)
Han CG Kemper (berthankemper@zonnet.nl)
Phuti PJ Makgae (pjmakgae@yahoo.com)

Version: 2 Date: 22 August 2009

Author's response to reviews: see over
22 August 2009

Dr Rikki Graham
Senior Assistant Editor
BMC-series journals

Dear Dr Graham

RE: MS:1305689340251359- Development and tracking of central patterns of subcutaneous fat of rural South African children aged 5 to 15 years: Ellisras Longitudinal Study

In response to the Reviewer’s suggestion, I resubmit the manuscript text for consideration by the BMC series Journals. The text covers the answers to all the Reviewer’s questions raised in the revision. Thank you for all the comments and judgment. The opinions were really helpful to eliminate misunderstanding and several major flows and to make the manuscript more accurate and valuable. I am grateful for the opportunity to reconsider our work for publication. I hope the revised manuscript will come up to the Reviewers expectations.

Kinds regards

Dr KD Monyeki